Author's response to reviews

Title: A new extract of the plant calendula officinalis produces a dual in vitro effect: cytotoxic anti-tumor activity and lymphocyte activation

Authors:

Eva E Jimenez-Medina (evamjm@fundacionhvn.org)
Angel Miguel AM Garcia (angel.miguel.exts@juntadeandalucia.es)
Laura L Paco (arualelche@telefonica.net)
Ignacio I Algarra (ialgarra@ujaen.es)
Antonia A Collado (antonia.collado.sspa@juntadeandalucia.es)
Federico F Garrido (federico.garrido.sspa@juntadeandalucia.es)

Version: 4 Date: 4 May 2006

Author's response to reviews:

Dear Editor,

This revised version, of our manuscript "A new extract of the plant calendula officinalis produces a dual in vitro effect: cytotoxic anti-tumor activity and lymphocyte activation" (MS: 7988736339546422), has following changes:

- The references have been corrected
- A new author has been included

Regards,

Angel Miguel Garcia